

JGR Biogeosciences



RESEARCH ARTICLE

10.1029/2021JG006304

Key Points:

- Over the twentieth century, the global maize and soybean yields driven by environmental and management factors increase
- Projected future [CO₂] increase has a positive effect, and warmer climate has a negative effect on soybean yield by the 2090s
- Projected maize yield is increased by 20% under RCP4.5-SSP2 but is decreased by 14% under RCP8.5-SSP5 by the end of the 21st century

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

A. K. Jain, jain1@illinois.edu

Citation:

Lin, T.-S., Song, Y., Lawrence, P., Kheshgi, H. S., & Jain, A. K. (2021). Worldwide maize and soybean yield response to environmental and management factors over the 20th and 21st centuries. *Journal of Geophysical Research: Biogeosciences*, *126*, e2021JG006304. https://doi. org/10.1029/2021JG006304

Received 23 FEB 2021 Accepted 13 OCT 2021

Author Contributions:

Conceptualization: Tzu-Shun Lin, Atul K. Jain

Data curation: Tzu-Shun Lin, Yang Song, Peter Lawrence, Atul K. Jain Formal analysis: Tzu-Shun Lin, Atul K. Jain

Funding acquisition: Atul K. Jain **Investigation:** Tzu-Shun Lin, Yang Song, Atul K. Jain

$\ensuremath{\textcircled{}^\circ}$ 2021. The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

Worldwide Maize and Soybean Yield Response to Environmental and Management Factors Over the 20th and 21st Centuries

Tzu-Shun Lin¹, Yang Song², Peter Lawrence³, Haroon S. Kheshgi⁴, and Atul K. Jain¹

¹Department of Atmospheric Sciences, University of Illinois, Urbana, IL, USA, ²Department of Hydrology and Atmospheric Sciences, University of Arizona, Tucson, AZ, USA, ³National Center for Atmospheric Research, Boulder, CO, USA, ⁴ExxonMobil Research and Engineering Company, Annandale, NJ, USA

Abstract A land process model, Integrated Science Assessment Model, is extended to simulate contemporary soybean and maize crop yields accurately and changes in yields over the period 1901-2100 driven by environmental factors (atmospheric CO, level ([CO₃]) and climate), and management factors (nitrogen input and irrigation). Over the twentieth century, each factor contributes to global yield increase; increasing nitrogen fertilization rates is the strongest driver for maize, and increasing [CO,] is the strongest for soybean. Over the 21st century, crop yields are projected under two future scenarios, RCP4.5-SSP2 and RCP8.5-SSP5; the warmer temperature drives yields lower, while rising [CO₂] drives yields higher. The adverse warmer temperature effect of maize and soybean is offset by other drivers, particularly the increase in [CO₃], and resultant changes in the phenological events due to climate change, particularly planting dates and harvesting times, by 2090s under both scenarios. Global yield for maize increases under RCP4.5-SSP2, which experiences continued growth in [CO₃] and higher nitrogen input rates. For soybean, yield increases at a similar rate. However, in RCP8.5-SSP5, maize yield declines because of greater climate warming, extreme heat stress conditions, and weaker nitrogen fertilization than RCP4.5-SSP2, particularly in tropical and subtropical regions, suggesting that application of advanced technologies, and stronger management practices, in addition to climate change mitigation, may be needed to intensify crop production over this century. The model also projects spatial variations in yields; notably, the higher temperatures in tropical and subtropical regions limit photosynthesis rates and reduce light interception, resulting in lower yields, particularly for soybean under RCP8.5-SSP5.

Plain Language Summary A land surface model is used to estimate changes in global maize and soybean yields in response to changes in environmental conditions (climate (temperature and precipitation) and carbon dioxide concentration ($[CO_2]$)), and agricultural management activities (irrigation, nitrogen application, and dynamic planting time decisions) over the 20th and 21st centuries. Estimated current crop yields compare well with the observed crop yields circa 2000. We then project how maize and soybean resources may change in the future under two climate and socio-economic assumptions: a high-end emission pathway (RCP8.5-SSP5) and a mitigated emission pathway (RCP4.5-SSP2). We find that future increase in $[CO_2]$ drives yield higher and alleviates the negative climate impacts on soybean productivity. The maize yield increases under the mitigated emission pathway because of earlier planting dates and the continued $[CO_2]$ growth and nitrogen application, the maize yield reduces by 14% by the end of this century, suggesting that climate change mitigation and improved agricultural technologies and practices may be needed to intensify crop production over this century. Crops' exposure to heat stress during grain formation is represented by canopy temperature and is projected to increase by the 2090s.

1. Introduction

Crop yield has and will be affected by environmental factors, such as atmospheric carbon dioxide level $([CO_2])$ and changes in temperature and precipitation patterns. For instance, an increase in the $[CO_2]$ positively influences the crop yields due to an increase in the photosynthesis process and reduced water requirement under increasing $[CO_2]$ (Ainsworth, 2008; Gray et al., 2016). On the other hand, increasing



Methodology: Tzu-Shun Lin, Yang Song, Peter Lawrence, Haroon S. Kheshgi, Atul K. Jain Project Administration: Atul K. Jain Software: Tzu-Shun Lin Supervision: Atul K. Jain Validation: Tzu-Shun Lin, Yang Song, Atul K. Jain Visualization: Tzu-Shun Lin, Peter Lawrence, Haroon S. Kheshgi Writing – original draft: Tzu-Shun Lin, Atul K. Jain Writing – review & editing: Tzu-Shun Lin, Yang Song, Peter Lawrence, Haroon S. Kheshgi, Atul K. Jain

temperature increases atmospheric water demand, causing water stress and drought that severely decrease crop growth and yields worldwide (Lobell et al., 2014; Rahimi-Moghaddam et al., 2018). The Global Gridded Crop Model Intercomparison (GGCMI) of Agricultural Model Intercomparison and Improvement Project (AgMIP) compares the crop yield results for global gridded crop models over this century for maize, wheat, rice, and soybean (Elliott et al., 2015; Müller et al., 2017). The AgMIP study generally finds that the models' estimated effects of environmental factors on crop yields are qualitatively consistent with the measurement studies (Müller et al., 2015).

However, the magnitude of these effects remains uncertain on a global scale and understanding of environmental factors' interactions with crop management practices (e.g., irrigation and nitrogen (N) fertilizer inputs), and the effects of extreme climate events, such as heat stress (HS) and droughts on yields, remain incomplete. Studies show that agricultural management practices generally can alleviate not only the adverse effects of climate change but also modulate the positive effects of increasing [CO2]. Levis et al. (2018) find irrigation can mitigate the yield losses due to increased respiration and water demand caused by temperature increases. Similarly, Rosenzweig et al. (2014) simulate higher yields under the assumption of no N-limitation. However, the AgMIP study does not explicitly study the synergistic effects of environmental and management factors and the effects of extreme climate events, specifically the effects of HS on crop productivity.

Studies show interactions between effects of environmental and management factors on crop yield. For instance, crop growth enhanced by the CO_2 fertilization effect is suppressed by low N supply and water stress conditions during the growing season (Ainsworth, 2008; Jain et al., 2009). Simultaneously, as the climate becomes warmer and wetter, inorganic soil N availability increases due to enhanced N mineralization associated with increased microbial decomposition and respiration rates (Jain et al., 2009; Rustad et al., 2001). Thus, improved N availability and uptake lead to enhanced crop productivity. Conversely, N deficits reduce crop growth rate, biomass, harvest index, and yields (Pandey et al., 2000). Although some crops are N fixation crops, for example, soybean field measurements suggest N fertilization can somewhat enhance yield depending on the yield potential of abiotic and biotic stresses (Cafaro La Menza et al., 2017; Salvagiotti et al., 2008).

Extreme climate events such as HS shorten both the vegetative and reproductive phases of the crop phenology (Asseng et al., 2004; Teixeira et al., 2013), resulting in the decrease of the average growing season length and crop yields. In addition, crops are sensitive to critical high temperatures during the reproductive period (Bolanos & Edmeades, 1996; Deryng et al., 2014; Gourdji et al., 2013). During this period, the negative impact of high temperature reduces carbon assimilation at the leaf level and directly impairs flower and crop grain development, resulting in crop yield loss (Sage et al., 2015; Teixeira et al., 2013).

This study aims to address three specific questions: (a) what are the synergistic effects of environmental $([CO_2] \text{ and climate})$ and management (irrigation and nitrogen input) factors on maize (a C4 crop) and soybean (a C3 crop) yields over the past century? (b) what are the effects of extreme HS on maize and soybean yields over the past century? (c) how do these effects change under two future scenarios over the 21st century?

It is to be noted that while several modeling studies have evaluated the effects of environmental factors, such as climate change, on crop yields at the site, country, and/or global scale, this is the first study, which quantifies not only the synergistic effects of climate change and management factors but also extreme climate events on crop yields over the past and 21st centuries. In addition, while few previous modeling studies have quantified the effects of HS, either at a regional or global scale, they use prescribed surface air temperature data to calculate the HS effect. In contrast, this is the first study that estimates the HS effect using the canopy temperature on a global scale because the canopy temperature explains better yield reductions associated with heat stress under drier, rainfed, and irrigated conditions.

We address the above three questions using a process-based land surface model, the Integrated Science Assessment Model (ISAM) (Song et al., 2013). The model is driven by historical climate data (1901–2015) and projected climate data (2016–2100) for two Representative Concentration Pathways: RCP4.5 and RCP8.5. The other model input variable–N input is developed here using the literature data for the historical time and two Shared Socioeconomic Pathways, SSP2 and SSP5, which represent agricultural activities under



RCP4.5 and RCP8.5 scenarios, respectively (O'Neill et al., 2014; van Vuuren et al., 2011). We also evaluate our model results for crop yield at regional (Figure S1 in Supporting Information S1) and global scale by comparing data and other published model studies for the historical and future periods.

2. Methods and Input Data

2.1. Model Description

ISAM, improved upon and used in this study, is a coupled biogeochemical and biogeophysical model with $0.5^{\circ} \times 0.5^{\circ}$ spatial resolution and multiple temporal resolutions ranging from half-hour to yearly time steps. It simulates C, N, energy, and water budgets for various terrestrial ecosystems through photosynthesis, surface hydrology, radiative transfer, carbon allocation, and ecosystem respiration (Barman et al., 2014a, 2014b; Yang et al., 2009). Moreover, ISAM incorporates crop growth processes for C3 and C4 food crops (maize, soybean, wheat, and rice) and bioenergy grasses (miscanthus, cave-in-rock, and alamo), which are evaluated at site-level, regional, and global scales (Gahlot et al., 2020; Lin et al., 2017; Niyogi et al., 2015; Song et al., 2013, 2015, 2016). Some of the important features, unique to ISAM and critical for crop yield calculations, include (a) dynamic crop-specific phenology and carbon allocation schemes (Song et al., 2013, 2015), accounting for the sensitivity of different crops to extreme environmental conditions; (b) dynamic vegetation structures, which better capture seasonal variability in leaf area index (LAI), canopy height, and root depth; (c) dynamic root distribution processes at the depth that improve simulated root-mediated soil water uptake and transpiration. The current study considers two crops: maize and soybean.

ISAM has been extensively calibrated, validated, and evaluated for agricultural applications (Gahlot et al., 2020; Niyogi et al., 2015; Song et al., 2013, 2015, 2016) and in nonagricultural application studies (Barman et al., 2014a, 2014b, 2016; El-Masri et al., 2013, 2015; Gahlot et al., 2017; Jain et al., 2006, 2013). Specifically, modeled estimated LAI, above and below-ground biomass, yield, and carbon, water, and energy fluxes for maize and soybean are calibrated and evaluated using the site level and county scale observation data (Song et al., 2013; Niyogi et al., 2015).

To improve the ISAM simulated crop growth processes and yields at a global scale, we extend ISAM to include (a) crop-specific dynamic planting time decision (Text S1 in Supporting Information S1), (b) crop-specific seeding rates (Text S2 in Supporting Information S1), (c) the curvature to the light response curve for the CO_2 fertilization effect (Text S3 in Supporting Information S1), (d) nutrient (i.e., N) stress while allocating the assimilated carbon to leaf, root, stem, and grain pools (Text S4 in Supporting Information S1), and (e) extreme heat stress effect during the crop reproductive stage of penology (Text S5 in Supporting Information S1). After implementation of newly added processes, modeled yields for the two crops are further evaluated for elevated $[CO_2]$, heat stress effect, N input, and irrigation effects using Free-Air Carbon dioxide Enrichment (FACE) experiments and other site-specific data sets, and published model results at specific sites, regional, and global scales over historical time and under future scenarios.

2.2. Input Data

2.2.1. Environment Forcing Data

Atmospheric CO_2 concentrations and climate conditions for historical (1901–2015) and future (2016–2100) periods are inputs for ISAM simulations of crop productivity. For the historical period, we use yearly $[CO_2]$ data from the Global Carbon Budget (Quéré et al., 2018) and climate forcing data from Climate Research Unit–National Centers for Environment Prediction (CRU-NCEP, Harris et al., 2014; Viovy, 2016), which are available at a 6-hr time scale. The future crop productivity calculations are performed for two climate scenarios: RCP 4.5-SSP2 and RCP8.5-SSP5 (O'Neill et al., 2014; van Vuuren et al., 2011). RCP4.5-SSP2 (hereafter referred to as RCP4.5) is a scenario to stabilize the total radiative forcing to 4.5 W/m² by 2100. In RCP4.5, economic, societal, and technological trends are assumed to be similar to historical patterns.

RCP8.5-SSP5 (hereafter referred to as RCP8.5) is a high energy demand and relatively rapid economic development scenario, causing high emissions and increased greenhouse gas (GHG) concentration by 2100. This scenario represents the highest GHG emission scenario available in the literature, leading to total radiative forcing to 8.5 W/m² by 2100. ISAM simulations for the two scenarios use $[CO_2]$ data from Meinshausen



et al. (2011) and climate forcing data from a single ensemble member of NCAR's Community Earth System Model (CESM) results (Levis et al., 2018; Ren et al., 2018). The CESM model results are bias-corrected using the CRU-NCEP climate data as described in Text S6 in Supporting Information S1. It is to note that the future input data, such as climate input data, is one primary but the inevitable source of uncertainty in modeling studies (Barman et al., 2014a, 2014b; Kheshgi et al., 1999; Jagtap & Jones, 2002). CESM climate scenarios that provide diurnal cycles of climate data allow us to assess the heat stress impact on maize and soybean yields at the hourly scale, which other global climate model outputs do not provide.

2.2.2. Crop Specific Area

The cropland area varies for the historical model simulations from 1901 to 2015. Thereafter, the area under the future scenarios remains fixed at an average of 1996–2005 conditions (Figure S2 in Supporting Information S1). The time-varying crop-specific annual area for the historical period is generated at $0.5^{\circ} \times 0.5^{\circ}$ from a combination of global crop-specific harvested areas circa year 2000 (M3, Monfreda et al., 2008) and the Land-Use Harmonization 2 data sets for the period 1901–2015 (LUH2, Hurtt et al., 2020). The process for calculating crop-specific areas at a 0.5×0.5 spatial scale is described in Text S7 in Supporting Information S1. Between 1901 and 2015, the maize area increased from 88 to 167 million ha, whereas the soybean area increased from 45 to 86 million ha (Figure S2 in Supporting Information S1). The spatial distribution of croplands for 1996–2005 shows the higher areas for both crops in midwest North America (NA), Europe (EU), South America (SA), South and Southeast Asia (SSEA), and the northeastern plain of China (CHN) (Figure S3 in Supporting Information S1).

2.2.3. Crop Specific N Input Amount

Crop-specific annual spatial distribution of N input (fertilizer, manure, and atmospheric deposition) rates (kg N/ha) are estimated for historical time and the two future scenarios using the method described in Text S8 in Supporting Information S1. We use the LUH2 data set for N fertilizer (Hurtt et al., 2020), Lamarque et al. (2011), and Tian et al. (2018) data for airborne nitrogen deposition (wet + dry), M3 total N input (sum of fertilizer, manure, and deposition) data for maize and soybean (Mueller et al., 2012), and manure data from Zhang et al. (2017).

The estimated global N fertilizer for maize over the period 1961–2010 was 509 Tg, which was consistent with the other studies' estimated value of 517 TgN (Ladha et al., 2016). In the year 2000, the total N input rate for maize and soybean were 108 kgN/ha and 59 kgN/ha (Figure S4a in Supporting Information S1). Of these totals, 77% and 64% of N are attributed to N fertilizer for maize and soybean. N manure contributes 15% for maize and 19% for soybean. Higher consumption appears in higher production regions, including CHN, NA, and EU (Figure S5 in Supporting Information S1). By the 2090s, N input rates for maize are about 200 kgN/ha under RCP4.5 and 107 kgN/ha under RCP8.5; and they are about 88 kgN/ha under RCP4.5 and 59 kgN/ha under RCP4.5 (Figure S5 in Supporting Information S1). N consumption is especially increased in AF and SA under RCP4.5 (Figure S5 in Supporting Information S1). The N input rates for soybean were almost 50% less than those of maize because it is an N-fixing crop. Some crop modeling studies assume that neither N fertilizer nor N stress is applied to soybean (Iizumi et al., 2017). In contrast, we follow field-based studies, which show N fertilizer is applied to soybean productivity.

The global N fertilizer rates increase for both crops over historical time (Figure S4b in Supporting Information S1). The global average N fertilizer application rates are higher under RCP4.5 (Figure S4b in Supporting Information S1) for future scenarios due to lower N use efficiency (Popp et al., 2017) and follow the historical trend. In contrast, rates decrease under RCP8.5 due to an increase in soil nitrogen uptake efficiency due to technological change (Popp et al., 2017).

The total manure N rates increased from 7.5 (maize) and 5.5 (soybean) kgN/ha in 1901 to 16.5 (maize) and 11.0 (soybean) kgN/ha in 2000 (Figure S4c in Supporting Information S1). Of these total, about 35% is assumed organic N and 65% is assumed inorganic (ammonium) N (Dangal et al., 2019). Manure application rate is increasing for maize but more in RCP4.5 than in RCP8.5 (Figure S4c in Supporting Information S1). The manure application rate for soybean also increases under both scenarios, but at a slower rate than maize and the rate of increase is similar in both scenarios.

Historical N deposition rate had been increased since 1940, and there was a near peak around 1990 followed by a decrease until 2015 (Figure S4d in Supporting Information S1). Future N deposition rates follow the fossil-fuel combustion patterns and increase under RCP8.5 but decrease under RCP4.5.

In ISAM, the N fertilizer, which is inorganic N, is added to the soil mineral N pools based on NH4⁺/NO3⁻ (Nishina et al., 2017) after emergence. The N fertilizer is evenly distributed for the first four weeks after emergence. The inorganic and organic manure N applications follow the same timing as N fertilizer, but the organic manure N is added to the soil, which is decomposed to mineral N gradually, and the inorganic N is added to the soil N mineral pool. N deposition has two components NHx and NOy, which are added to soil NH4⁺ and NO₃⁻ pools at a weekly interval.

2.2.4. Irrigation

The cropland area of each grid cell is divided into irrigated and rainfed areas. The irrigated fractional area for each grid cell for 1901–2015 is assigned based on LUH2 (Hurtt et al., 2020) (Text S7 in Supporting Information S1). After that, the irrigated area in each grid cell under the future scenarios remains fixed at the 1996–2005 level. ISAM provides water on an irrigated fraction of land when the root-zone soil water is limiting for crop photosynthesis, but the crop LAI is greater than zero as described in Text S9 in Supporting Information S1.

Irrigation fraction continuously increased in the twentieth century but is less than 20% of individual crop areas (Figure S6a in Supporting Information S1). Regarding irrigation water amount, ISAM estimates approximately 34 and 16 km³ in the year 2000, due to growing maize and soybean on irrigated croplands, which fall within global gridded crop models' (11 models for maize and 10 models for soybean) estimated range values (11–237 for maize and 3–36 for soybean) (AgMIP, Müller et al., 2019) (Figure S6b in Supporting Information S1).

2.3. Historical Yield Date For Model Evaluation

To evaluate the model estimated yields with the data available in the literature (Section 3.1), we use three global gridded data sets: GDHY (Iizumi et al., 2014), M3 (Monfreda et al., 2008), and SPAM (You et al., 2014); and one country scale data set: FAOstat (FAOstat, 2017). These data sets provide yield for different periods, GDHY for 1982–2006, M3 for ca. 2000 (mean for 1997–2003), SPAM for 2000 and 2005, and FAOstat for 1961–2018. We selected a 10-year period, 1996–2005, to compare the model results with the literature data. There are a total of 23 values available from the literature data over this period, one (2000) from M3, ten (1996–2005) each from GDHY and FAOstat, and two (2000 and 2005) from SPAM. We then select 23 model simulated values, which correspond to the same years. We calculate mean and median values from modeled and literature data values, which we compare in Section 3.1.

2.4. Experimental Design and Analysis

ISAM is spun-up by repeating the hourly climate forcing data (Harris et al., 2014; Viovy, 2016) for the period 1901–1920, and fixed the year 1900 values for $[CO_2]$ (296.8 ppm, Le Quéré et al., 2018), crop area, and N input (Section 2.2). The model is run until the soil temperature, moisture, and C and N pools reach a steady state. The spin-up process is described in detail in El Masri et al. (2015) and Song et al. (2016). The spin-up simulation follows transient model simulations with prescribed spatial and temporal forcing data for climate, N input, and $[CO_2]$ for 1901–2100. The model calculates the annual yields and corresponding C, N, water, and energy fluxes. Here, simulated crop yields are converted from modeled dry grain carbon with the assumption of 45% carbon in the dry matter (Lobell et al., 2002) and average water content of 12% for maize and 9% for soybean (Wirsenius, 2000).

We also generate three sets of simulations: one to examine crop productivity over the period 1901–2005, and the other two for the two scenarios over the period 2006–2100. Each set consists of four model experiments (E_{CO2} , E_{Cli} , E_{Nit} , and E_{Irr}) to examine each factors' effects, and one additional experiment to study the total effect of management (E_{Man}) over the two time periods (Table 1). In the reference case (E_{Ref}), all four factors vary with time over two time periods. The five additional simulations are performed differently for the periods 1901–2005 and 2006–2100. For the 1901–2005 set, one of the four factors remains fixed at the 1900



Table 1

Model Experiment Design to Study the Effects of Individual Environmental and Management Factors Over 1901–2100

Cases	[CO ₂]	Climate	N Input	Irrigation	Heat stress (HS)
E _{Ref}	1901–2100	1901–2100	1901–2100	Irrigation on irrigated land, rainfed conditions in rainfed land	w/heat stress calculated using canopy temperature
E _{CO2} ^a	Fixed at 1900 level (296.8 ppm) for 1901–2005 and fixed at 2005 level (378.2 ppm) for 2006–2100	Same as in E_{Ref}	Same as in $\mathrm{E}_{\mathrm{Ref}}$	Same as in $\mathrm{E_{Ref}}$	Same as in $\mathrm{E}_{\mathrm{Ref}}$
E _{Cli} ^a	Same as in $\mathrm{E}_{\mathrm{Ref}}$	Recycle 1901–1920 for 1901–2005 and recycle 1996–2005 for 2006-2100	Same as in E _{Ref}	Same as in $\mathrm{E}_{\mathrm{Ref}}$	Same as in $\mathrm{E}_{\mathrm{Ref}}$
$E_{Nit}^{ a}$	Same as in $\mathrm{E}_{\mathrm{Ref}}$	Same as in $\mathrm{E}_{_{\mathrm{Ref}}}$	No N input application	Same as in $\mathrm{E_{Ref}}$	Same as in $\mathrm{E}_{\mathrm{Ref}}$
$E_{Irr}^{\ a}$	Same as in $\mathrm{E}_{\mathrm{Ref}}$	Same as in $\mathrm{E}_{\mathrm{Ref}}$	Same as in $\mathrm{E}_{\mathrm{Ref}}$	No irrigation on irrigated and rainfed lands	Same as in $\mathrm{E}_{\mathrm{Ref}}$
E _{Man}	Same as in $\mathrm{E}_{\mathrm{Ref}}$	Same as in $\mathrm{E}_{\mathrm{Ref}}$	No N input application	No irrigation on irrigated and rainfed lands	Same as in $\mathrm{E}_{\mathrm{Ref}}$
E_{w/o_HS}	Same as in E _{Ref}	Same as in E _{Ref}	Same as in $\mathrm{E}_{\mathrm{Ref}}$	Same as in E _{Ref}	w/o heat stress effect
$E_{w/_HSAT}$	Same as in $\mathrm{E}_{\mathrm{Ref}}$	Same as in $\mathrm{E}_{\mathrm{Ref}}$	Same as in $\mathrm{E}_{\mathrm{Ref}}$	Same as in $\mathrm{E}_{\mathrm{Ref}}$	w/heat stress calculated using air temperature

^aWe first run the model of the E_{Ref} case through the period 1901–2005. Then conduct environmental and management factor experimental simulations for the period 2006–2100.

level, whereas all other factors vary with time as in E_{Ref} . For the 2006–2100 sets, we first continue to run the model for the E_{Ref} case through the period 2006–2100. Next, we run the model from 2006 to 2100, with one of the four factors remaining fixed at the 2005 level, but other variable values are assumed to change as in the E_{Ref} case. In the E_{Nit} and E_{Irr} experiments, the N input or irrigation input is assumed to be zero, respectively. In the E_{Man} experiments, both N input and irrigation are assumed to be zero. We then estimate the effect of each factor by differencing the yields between the reference case and one of the five simulations: CO_2 fertilization ($E_{Ref}-E_{CO2}$), climate ($E_{Ref}-E_{Cli}$), irrigation ($E_{Ref}-E_{Irr}$), N input ($E_{Ref}-E_{Nit}$), and combined N input and irrigation ($E_{Ref}-E_{Man}$).

To identify and evaluate HS during reproductive stages, we performed two additional experiments. The first experiment is $E_{w/o_HS'}$ which does not account for the HS effect. The second experiment, $E_{w/_HSAT'}$ accounts for the HS effect, but the effect is calculated using air rather than canopy temperature (Table 1). The effect of HS with canopy temperature and air temperature is calculated by subtracting E_{w/o_HS} and $E_{w/_HSAT}$ from $E_{Ref'}$.

For 1901–2005, we calculate the contribution of individual factors (in %) over a given time relative to the E_{Ref} case (e.g., both averaged over 1996–2005). For the two future scenarios, we compare the results for the 2090s (e.g., averaged over 2090–2099) relative to 1996–2005. The results are masked out using the irrigated and rainfed areas for each simulation at 0.5° x 0.5° (latitude x longitudes) spatial resolution. The total yield for each grid-cell is calculated by combining the weighted irrigated and rainfed yield (Equation S22 in Supporting Information S1). The results are presented at a spatial scale at 0.5° x 0.5° and a regional scale. To obtain the model results at regional scales, we average the spatial results for each crop over its cropland in six regions shown: NA, SA, EU, AF, CHN, and SSEA (Figure S1 in Supporting Information S1). It is to be noted here that the model simulates yield and production for all crop areas available in the literature. However, spatial plots do not display the yields and production for those grid cells that contain less than 0.1% of maize or soybean crop areas, such as some Canadian Prairies locations.

3. Results

3.1. Model Evaluation

Here, we evaluate the model performance for those processes, which are added in this study.



Table 2

Global and Regional-Scale Percent Bias (PBIAS, %) for Maize and Soybean Yields From the Original Version (Song et al., 2013), Revised Versions (This Study, E_{Ref}) of ISAM, and AgMIP Models Compared to Various Literature Data for 1996–2005^a

	Maize			Soybean				
	PBIAS (%)			Yield (t/ha)	PBIAS (%)			Yield (t/ha)
Global/Region	Song et al. (2013) (N = 23)	This study ^b $(N = 23)$	$\begin{array}{c} \text{AgMIP}^{\text{d}} \\ (N = 276) \end{array}$	Literature data $(N = 23)$	Song et al. (2013) ($N = 23$)	This study ^c $(N = 23)$	$\begin{array}{c} \text{AgMIP}^{\text{d}} \\ (N = 276) \end{array}$	Literature data $(N = 23)$
Global	-20.7	9.5	-21.0	4.6	-30.5	4.9	-10.4	2.2
North America (NA)	-13.8	11.2	9.4	8.6	-24.2	-2.5	2.8	2.6
South America (SA)	-44.2	4.0	-86.6	3.1	-13.6	30.0	-15.3	2.4
Europe (EU)	-21.8	7.3	18.9	6.0	-4.6	15.7	27.1	3.0
Africa (AF)	-75.0	-4.3	-132.6	1.8	-94.5	2.4	-75.6	1.0
China (CHN)	-14.1	-7.4	-14.2	5.0	-81.6	-43.0	-31.7	1.8
South and South East Asia (SSEA)	-15.5	19.1	-96.4	2.2	-110.5	3.4	-80.7	1.0

Note. N is the number of values used to calculate PBIAS.

^aThe literature data set are Iizumi et al. (2014) for the period 1996–2005, Monfreda et al. (2008, M3) for the year 2000, You et al. (2014, MapSPAM2000 & 2005) for the years 2000 and 2005, and FAOstat (2017) for period 1996–2005. So, there are N = 23 sample literature data values (see Section 2.3). ^bThe Original and Revised columns are the % bias (PBIAS) for w/o and w/ N stress effect on carbon allocation for maize (Text S4 in Supporting Information S1) and heat stress impact (Text S5 in Supporting Information S1). ^cThe Original and Revised columns are the % bias (PBIAS) for w/o and w/ N stress effect on carbon allocation (Text S4 in Supporting Information S1), seeding rates (Text S2 in Supporting Information S1), heat stress impact (Text S5 in Supporting Information S1), and revised elevated CO₂ effect (Text S3 in Supporting Information S1) for soybean. ^dThe AgMIP results are across 12 different crop models driven by AgMERRA with the default setting (Müller et al., 2019). For each model, 23 values are selected corresponding to the same years as literature data. So, there are N = 276 (=12 × 23) sample data values for the AgMIP case.

3.1.1. Crop Specific Planting Time and Seeding Rates

ISAM estimates crop-specific dynamic planting time decisions rather than fixed planting time at a gridscale (Text S1 in Supporting Information S1). So, the dates can vary with time, depending upon the changes in the environmental conditions. Here, we evaluate the model estimated planting time using the literature data (Elliott et al., 2015) in Figure S7 in Supporting Information S1. ISAM results show that estimated planting time is controlled by climate and soil conditions (Table S1 in Supporting Information S1) associated with crop-specific phenology. In addition, we update the crop seeding rates and residue amount (Text S2 in Supporting Information S1), which vary with planting conditions (Table S2 in Supporting Information S1). The updated seeding rates at the sowing time are usually lower for soybean in CHN, AF, and SSEA. After implementing these modifications, modeled soybean yields are reduced. The revised yields in these regions for the 1996–2005 average compare better with the literature data. The percent bias (PBIAS, Text S10 in Supporting Information S1) of the modeled yield is reduced (Table 2).

3.1.2. CO₂ Fertilization Effect

Crop productivity under elevated $[CO_2]$ from the original version of ISAM compared with measurements at FACE sites (Table S3 and Text S3 in Supporting Information S1 for FACE site calculations) shows that while the modeled yield for maize is consistent with FACE site measurements (Table S4 in Supporting Information S1), it is overestimated for soybean (Figure S8a and Table S4 in Supporting Information S1). This overestimation is because the electron transport rate calculations did not account for the curvature of the light response curve, resulting in the overestimation of photosynthesis, canopy temperature, and stomatal conductance (Figures S8b and S8c in Supporting Information S1). However, after implementing curvature to the light response curve (see the detailed description of the method and results in Text S3 in Supporting Information S1), the revised ISAM results under irrigated conditions are consistent with measured values. While the range of AgMIP model results under irrigated conditions for maize is comparable to the FACE experiment results but unable to calculate the reduction in maize yield. However, ISAM can calculate maize yield reduction as observed in the FACE experiment (Table S4 in Supporting Information S1).



3.1.3. N Fertilization Effect on Maize Yield

The ISAM model of Song et al. (2013) also overestimated the maize yields at the lower N application rates (Figure S9a in Supporting Information S1) because the model overestimates the carbon allocated to the grain formation under the N stress conditions (i.e., the ratio of N supply and N demand). However, model results for soybean are consistent with the measured data (Figure S9b in Supporting Information S1). After accounting the N stress effect on the carbon allocation to grain during initial and post-reproductive (grain-filling) stages of phenology (Text S4 in Supporting Information S1), the revised ISAM results for a lower N fertilizer rate at six sites (Table S5 in Supporting Information S1) show stronger N stress, and lower grain formation compared to the results estimated based on the original ISAM (Figure S9a in Supporting Information S1). These results, which are consistent with the field experiment studies, suggest that maize growth slows down at the lower N supply rates, causing a decline in yield (Alemayehu et al., 2015; Gehl et al., 2005; Getachew & Belete, 2013; Hammad et al., 2011). The revised model also improves the over-estimation of simulated global and regional maize yields, particularly in AF, where N is a limiting factor (Table 2).

3.1.4. Heat Stress (HS) Effect

We implement HS in ISAM by accounting for the impact of HS on reducing carbon allocation during the reproductive phases of the phenology (Text S5 in Supporting Information S1). While other studies use air temperature for the calculation of the HS effect (Deryng et al., 2014; Teixeira et al., 2013), here, our model simulations consider the crop canopy temperature (Text S5 in Supporting Information S1), which is shown to explain better yield reductions associated with HS (e.g., Gabaldón-Leal et al., 2016; Siebert et al., 2014; Webber et al., 2017). To evaluate the model estimated yield performance with HS effect, we compared the three cases (E_{Ref} , $E_{\text{w/o, HS}}$, and $E_{\text{w/ HSAT}}$) of correlation coefficients of 1982–2006 period detrended yields for ISAM and FAOstat (2017) (Table S6 and Figure S10 in Supporting Information S1). The results show that the ISAM simulated temporal variability for maize yields at global and regional scales is somewhat better in E_{Ref} than in $E_{W/0 HS}$. While the improvement in reproducing the observed temporal variability of maize yield occurs in all regions except for SA (Table S6 in Supporting Information S1), the differences between the two cases are small. The impact of HS on maize yield is more in AF and SSEA, where the temperature is higher than in other areas. Although maize growth also responds to HS calculated using air temperature, the resulting performance in capturing observed temporal variability in $E_{w/HSAT}$ is less than in E_{Ref} (Table S6 in Supporting Information S1). However, HS shows negligible effects on simulated soybean yield variability because of soybean's higher critical temperature (Text S5 in Supporting Information S1). Gourdji et al. (2013) showed that exposure to high critical temperatures in the reproductive period could happen in many places for maize but not for soybean in the twentieth century.

3.1.5. Percent Biases Results for ISAM

Overall, the PBIAS results show the model estimated yields at global and regional scales are compared well with the observations after model improvements, except for maize in SSEA (19%), soybean in CHN (-43%), and SA (30%) (Table 2). The PBIAS for maize yield is reduced from -21% to 10%, and for soybean from -31% to 5% globally. On the regional scale, modeled maize yield improves in NA, SA, EU, CHN, and AF, and soybean yield in NA, CHN, AF, and SSEA (Table 2). The remaining model biases might be due to the model limitations in estimating nutrients' limitation, crop mortality effects due to ozone, wind, hail, weeds, pests, disease, and/or due to not accounting for the cropping systems in the model. Also, uncertainty in the input data, such as climate, soil, or crop management, might have introduced the biases in the modeled yield (Barman et al., 2014a, 2014b; Kheshgi et al., 1999; Jagtap & Jones, 2002), which we plan to carry out in our future modeling analysis.

3.2. Model Estimated Crop Yields and Productions

3.2.1. Yields and Productions for the Historical Time

ISAM results for maize and soybean yields for the period 1996–2005 are compared with global data sets available in the literature (see Section 2.3) in Figure 1, and production on a $0.5^{\circ} \times 0.5^{\circ}$ grid in Figure 2. Overall, the calculated global maize yield averaged from literature data sets is 4.6 t/ha (Table 2), with NA





Figure 1. Global and regional-scale comparisons of (a) maize and (b) soybean annual yield (t/ha) from Integrated Science Assessment Model (ISAM), Agricultural Model Intercomparison and Improvement Project (AgMIP) (Müller et al., 2019), and the available data set values in the literature. Model results are the original (ISAM_O) and revised (ISAM_R) versions for the period 1996–2005, and the literature data include Iizumi et al. (2014) for years 1996–2005, Monfreda et al. (2008, M3) for 2000, You et al. (2014) for the years 2000 and 2005, and FAOstat (FAOstat, 2017) for years 1996–2005. The AgMIP results are values across12 different crop models driven by AgMERRA with the default setting (Müller et al., 2019). For each model, 23 values are selected corresponding to the literature data years values. The boxes are the interquartile ranges, the horizontal lines plotted in the boxes are the median values, and the whiskers indicate the highest and lowest values of the results. The green triangles marked in the boxes are the mean values.



Figure 2. The spatial distribution of annual maize and soybean production estimated by ISAM averaged over the same years as literature data are compared to an average of literature data. The distribution is expressed in tonnes of annual production for each $0.5^{\circ} \times 0.5^{\circ}$ grid-cell; the legend is, therefore, in units of tonnes/yr. The maize and soybean estimates from the Integrated Science Assessment Model (ISAM) model are weighted by irrigated and rainfed areas. The areas for both crops are masked by crop-specific areas. The literature data set is the average of three gridded products—Iizumi et al. (2014) for the period 1996–2005, Monfreda et al. (2008) for the year 2000, and You et al. (2014) for the year 2000 and 2005—as described in Section 2.3. The color scale is plotted on a logarithmic scale.

(8.6 t/ha) and AF (1.8 t/ha) are the highest and lowest yield regions. In contrast, global soybean yield is about half of the maize yield (2.2 t/ha), and regional values range between 1.0 t/ha in AF and SSEA and 3.0 in the EU (Table 2).

ISAM results for higher maize yield regions (NA, EU, and CHN) and lower maize yield regions (AF, SA, and SSEA) are within the range of observed contemporary yields (Figure 1a). ISAM also reproduces the measured pattern of soybean yields (Figure 1b) across high yield regions (NA, SA, and EU) and low yield regions (e.g., SSEA). Moreover, ISAM appears to capture the maize and soybean breadbaskets, including the Corn Belt in the United States, Argentina (Buenos Aires, Entre Rios, Cordoba, and Santa Fe), Brazil (Parana, Goias, Mato Grosso, and Minas Gerais), India (Madhya Pradesh, Maharashtra, and Rajasthan), and northeastern CHN (Figure 2). As also shown and discussed in the PBIAS section, the agreement between ISAM results and literature data is improved in the revised version of ISAM (ISAM_R), compared to the version of ISAM developed by Song et al. (2013), which is referred to as the original version of ISAM (ISAM_O) in Figure 1.

While ISAM captures similar patterns of literature data for maize and soybean productions (Figure 2) with higher production in grid cells with a higher cultivated area (Figure S3 in Supporting Information S1), some mismatches between modeled and literature spatial distribution data for harvested areas may have introduced some differences in production patterns. In general, the model underestimates maize production in the Corn Belt of the United States and north CHN, soybean production in Argentina, Brazil, and overestimates soybean production in northeastern CHN, and both production in Russia (Figure 2). Given that harvested crops' quantity and spatial patterns differ among the three literature gridded products applied here, the uncertainty for aggregated literature data values increases at the grid cell level (Porwollik et al., 2017). Therefore, it is perhaps not surprising that modeled production and yields of maize and soybean for some regions exhibit some differences from literature data.

3.2.1.1. Modeled Versus Measured Detrended Yield For 1982–2006

In addition, ISAM is able to reproduce the observed (FAOstat, 2017) detrended global and regional yields (Text S11 in Supporting Information S1 describes the method to calculate detrended yield) over the period 1982–2006 with the correlation coefficient, *r*, 0.7 for maize and 0.6 for soybean (Table S6 and Figure S10 in Supporting Information S1). These values are close to the middle of the range of values estimated based on the ensemble of global AgMIP model results (14 different model results for maize yield and 13 models results for soybean yield); 0.26–0.89 for maize and 0.00–0.64 for soybean (Müller et al., 2017) although crop areas and climate forcing are different. The model estimated detrended yields for both crops at the regional scale are also compared well with FAOstat (2017), except the values of *r* for soybean in AF, SSEA, and CHN and for maize in SSEA where ISAM estimated detrended values are higher than FAOstat (2017). The studies suggested that this might be related to the reporting year issue; some crop yields harvested at the end of the calendar year are reported by FAOstat (2017) in the following year report. Therefore, the detrended FAOstat yield might have a one-year delay in contrast to ISAM values in some years.

3.2.1.2. Comparison With AgMIP Model Results

Compared to literature data, 12 AgMIP crop model results for yields varied widely. However, an ensemble of the 12 model results compared better to literature data than the individual model results. Although the median of all AgMIP models gives better performance than the literature data (Figure 1), the estimated PBIASs for global and regional cases are higher than ISAM (Table 2), except for NA for maize and SA and CHN for soybean. The AgMIP models' simulated maize and soybean yields are out of the maximum range value reported by literature data in AF (PBIAS of -133% for maize and -76% for soybean) and SSEA (PBIAS of -96% for maize and -81% for soybean), where the actual productivity is low. Overestimating and underestimating crop yields from AgMIP models in these and other regions may be due to some processes and factors not considered in AgMIP models at the global scales, such as management practices, seeding rates, residue amount, and nutrient application (Müller et al., 2017).

3.2.2. Crop Yield Under RCP4.5 and RCP8.5 Scenarios

ISAM-estimated global crop yield changes are driven by environmental and management factors specified by scenarios RCP4.5 and RCP8.5. Changes in yield in the 2090s relative to that in 1996–2005 are shown in





Figure 3. (a) Maize and (b) soybean yield changes (%) at regional and global scales for the 2090s relative to the 1996–2005 average under RCP 4.5 (green bars) and RCP 8.5 (brown bars) scenarios. Solid bars are results for E_{Ref} and crosshatched bars are results for E_{Man} .

Figure 3 and spatial maps in Figure S11 in Supporting Information S1. For scenario RCP8.5, the estimated maize yield for E_{Man} case is projected to decrease across all regions except for the EU. For scenario RCP4.5, maize yield is projected to increase in all regions, except for CHN (Figure 3a). In contrast, soybean yield increases across all regions under both scenarios (Figure 3b).

3.3. The Effects of Changes in Environmental and Management Factors on Maize and Soybean Yields

Each of the four environmental and management factors (CO_2 , climate, N input, and irrigation) considered results in an estimated increase in maize and soybean yield at the global scale from 1901 to 1996–2005 (Table S7 for numbers and Figure S12a for maps in Supporting Information S1). The yields for both crops increase across all regions due to the CO_2 fertilization effect. However, the increase is stronger for soybean than for maize because for soybean, which is a C3 crop, photosynthesis is relatively less saturated under ambient [CO_2] (McGrath & Lobell, 2013). Without the [CO_2] increase, the global maize and soybean yields would have been lower by 5% and 19%, respectively (Table S7 in Supporting Information S1). Over the last century, climate change had a small positive global effect (1% and 4%), with some regions showing positive and negative effects (Figure S12a in Supporting Information S1).





Figure 4. Maize and soybean yield contributions (%) for the average over the 2090s (2090–2099) at global and regional scales because of (CO_2) , climate, N input, irrigation, and combined N input and irrigation under RCP4.5 and RCP8.5 scenarios (E_{Ref} in 2090s minus E_{XXX} in 2090s then divided by E_{Ref} in 1996–2005). E_{XXX} are the factor experiments shown in Table 1.

Out of the four factors studied here, N input increases the future maize yields, and $[CO_2]$ affects the soybean yield the most under the two future scenarios (Table S7 in Supporting Information S1 and Figure 4). On the other hand, climate decreases the yields for both crops under the two scenarios. Irrigation shows a slightly positive effect across all regions.

Our results also reveal that management factors modulate the $[CO_2]$ and climate effects on productivity for both crops. The increasing $[CO_2]$ prompts higher crop yield but simultaneously requires more N, exacerbating N limitation. At the same time, although elevated $[CO_2]$ promotes water use efficiency for crops, water is essential to maintain the increase in carbon assimilation and leaf area index, particularly for soybean. By implementing N and irrigation, crop yields show a higher response to elevated $[CO_2]$ (Figure S13 in Supporting Information S1). These model results are supported by the experimental results (Gray et al., 2016; Kim et al., 2006; Ruiz-Vera et al., 2015). In general, higher response occurs in the regions where both effects



of $[CO_2]$ and management on crop yield are higher (Figures S12b and S12c in Supporting Information S1). Therefore, the synergistic effect of $[CO_2]$ and management on crop yield is more positive in NA, EU, CHN for maize, and CHN and SSEA for soybean than in other locations (Figure S13 in Supporting Information S1). The effect is substantially higher in RCP8.5 than in RCP4.5.

Similarly, compared to climate change only effect, that is, without management case, interactive effects of management factors and climate change effect on crop substantially decreases/enhances crop productivity losses/gains in the 2090s (Figure S13 in Supporting Information S1). The N and water demand for crops increases under warmer and drier conditions. The demand also increases when the crop productivity is higher due to favorable environmental conditions. Therefore, crop growth shows a higher response in the 2090s than in recent years to management factors under both climate change scenarios.

4. Discussion

4.1. Model Evaluation Using Data

Global and regional crop yields estimated with the ISAM land surface model for the C3 crop soybean and the C4 crop maize are consistent with literature data averaged over 1996–2005 (Figures 1 and 2). IS-AM-calculated yield variability at regional and global scales over time, 1982–2006, is consistent with FA-Ostat (2017) (Figure S10 in Supporting Information S1). In addition to this study, the overall confidence in ISAM estimated yields for maize and soybean is strengthened by validating ISAM results at the site level (Song et al., 2013) and the county level (Niyogi et al., 2015).

4.2. Estimated Effects of Environmental and Management Factors

ISAM results show that over the past century and for two future scenarios, RCP4.5 and RCP8.5, environmental and management factors affect maize and soybean yields:

4.2.1. CO₂ Fertilization

We find that the modeled CO_2 fertilization effect is stronger for soybean across all regions than for maize, because ISAM's net photosynthesis rate increases due to higher carboxylation rates and lower photorespiration rates for soybean than for maize. The effect is stronger in tropical regions (SA, AF, and SSEA) (Figure 4, Table S7 and Figure S12 in Supporting Information S1) because of (a) greater availability of N via N fixing bacteria (seen in measurements in Ainsworth et al., 2002), (b) smaller LAI (Figure S14 in Supporting Information S1), which absorbs higher photosynthetically active radiation (PAR), because PAR is inversely proportional to LAI (seen in measurements and model results in Sakurai et al., 2014), and (c) higher temperature enhances the CO_2 fertilization effect on net photosynthesis rate because both the specificity factor of Rubisco for CO_2 and solubility of CO_2 in water decline relative to O_2 (seen in measurements, Bernacchi et al., 2006; Ruiz-Vera et al., 2013) with rising temperature.

In contrast, the CO_2 fertilization effect is higher for maize in temperate drier regions (e.g., parts of NA, EU, AF, and CHN) in 1996–2005 (Table S7 and Figure S12a in Supporting Information S1). Rising $[CO_2]$ can increase crop water productivity (the ratio of yield to evapotranspiration) due to the reduction in stomatal conductance, which indirectly enhances crop yield by ameliorating soil water stress in dry soils (seen in measurements in Leakey et al., 2006). Similar to the historical case, the yield increase is stronger for soybean than for maize under both future scenarios due to the CO_2 fertilization effect at regional and global scales in the 2090s (Figure 4, Figures S12b and S12c in Supporting Information S1 for maps).

We compared ISAM-estimated changes of global-average yield in 2080 for both crops with (w/) and without (w/o) CO_2 cases under RCP8.5 with model results available in the literature–AgMIP (Deryng et al., 2016) and NCAR's Community Land Model (CLM, Ren et al., 2018)—in Table 3. While ISAM-estimated maize and soybean yields for w/ CO_2 case falls within the interquartile range values of AgMIP model results, there are differences in the spatial patterns (Figure S15 in Supporting Information S1). This may be because the representation of management practices such as irrigation, crop residue, phenology, planting dates, and fertilizer application are different among AgMIP models (Deryng et al., 2016). For example, only three AgMIP models apply N fertilizer. Moreover, two models allow adaptation of planting time window (time



Table 3

Comparison of ISAM Estimated Global Maize and Soybean Yields Change to That of Other Model Results Available in the Literature: Percent Change in 2076–2085 (or 2080) Average Yield Relative to 1996–2005 (or 2000) Under the RCP8.5 Scenario

	ISAM		CLM ^a		AgMIP ^b		
Crop	w/ CO ₂	w/o CO ₂	w/ CO ₂	w/o CO ₂	w/ CO ₂	w/o CO ₂	
Maize	-3.8	-10.1	0.7	-9.2	[-16.4; 1.0]	[-28.2; -13.3]	
Soybean	25.1	-12.9	18.6	-9.6	[-12.1; 33.3]	[-40.5; -27.7]	

Note. The results for reference w/ CO2 (ERef, Table 1) and w/o CO2 (ECO2, Table 1) cases are compared. The nitrogen input is applied as per the E_{Ref} experiments. The ISAM yield results are weighted by fixed irrigated and rainfed areas based on the MIRCA2000 data set, which are the same as other model results.

^aResults are taken from Ren et al. (2018). N fertilization application is set at North American crop-specific levels everywhere and fixed over time. Then, yield is adjusted with nitrogen fertilizer assumptions based on FAO data. ^bResults are the interquartile range across all six global gridded crop models run with climate data from five global climate models (Deryng et al., 2016). The EPIC, GEPIC, and pDSSAT models apply fertilizer dynamically through the crop growing season: application occurs at specific stages of the crop development to consider the role of both application quantity and timing. PEGASUS applies fertilizer as a daily stress function and thus does not simulate the effect of fertilizer directly. LPJmL and LPJ-GUESS do not represent fertilizer application. PEGASUS and GEPIC allow for adaptation for variation of planting window with time, whereas the other GGCMs assumed fixed planting window overtime. interval between the earliest and latest possible planting dates), while the other four models assume planting window to remain fixed with time at the present dates. By considering the climatic effect on planting time, crops in ISAM are planted earlier when the temperature is higher, which can alleviate high-temperature exposure through the growing season. As such, our global crop yield w/o CO_2 case is higher than AgMIP but is similar to CLM results that use the same CESM climate forcing (Table 3).

4.2.2. Climate Change (Excluding Heat Stress)

The climate effect on yield over the past century differs by region (Table S7 and Figure S12a in Supporting Information S1), but is in line with previous studies (e.g., Challinor et al., 2014; Fodor et al., 2017; Rosenzweig et al., 2014). The hotter temperature over most of the tropical and subtropical regions reduced yields for both crops, even though these regions experience higher precipitation. In contrast, increased temperature enhanced yields for both crops in colder regions (NA, eastern EU, northeastern CHN, and in boreal latitude zones) (Figure S12a in Supporting Information S1) in 1996–2005, where moderate warming increased the length of the growing period. For future scenarios, the adverse effect of climate lowers yields of both crops in all regions, with a stronger effect on soybean than on maize (Figure 4), because the optimum leaf temperature for photosynthesis is higher for maize than that for soybean. For example, net photosynthesis rate, A, for maize (soybean) increases up to 50 μ mol CO₂/m²/s (25 μ mol CO₂/m²/s) with the leaf temperature that increases to 40°C (25°C); after that, A decreases with increasing leaf temperatures (Figure S16 in Supporting Information S1). Moreover, rising temperature increases crop respiration and reduces carbon use efficiency (CUE), defined as the ratio of net primary production to gross primary production (Zhang et al., 2013). Since soybean has lower CUE and shal-

lower rooting depths than maize (Yamaguchi, 1978), soybean incurs relatively higher carbon losses through respiration, resulting in a much stronger response to climate change (Figure 4). CLM had also discovered a more negative impact on soybean than maize yield (Lombardozzi et al., 2018).

4.2.3. Heat Stress

ISAM estimated results show a higher global HS effect during the reproductive stages on maize yield (-2.4%) and -5.9% under RCP4.5 and RCP8.5) than on soybean yield (-1.3% and -4.6% under RCP4.5 and RCP8.5) by the 2090s (Figure 5). While HS effect on yield is estimated in previous modeling studies using surface air temperature (e.g., Deryng et al., 2014; Teixeira et al., 2013), ISAM simulations are performed using canopy temperature rather than the surface air temperature because canopy temperature is warmer than the surface air temperature under drier and rainfed conditions. However, the canopy temperature is colder than the surface air temperature under irrigated conditions. So, the canopy temperature can ameliorate crop yield losses due to HS through irrigation. For example, the cooling effect of irrigation can decrease yield losses. The published global crop modeling studies, on the other hand, prescribed relatively warm air temperature (as opposed to canopy temperature while irrigating) for temperature and HS calculations. The model estimated heat stress-related agricultural hotspot regions are consistent with the published studies (Gourdji et al., 2013; Teixeira et al., 2013), which shows subtropical and temperate crop areas have large cropping areas under higher HS risk. ISAM results show that South Asia, Sahel, Eastern China, Spain, parts of Central Asia (e.g., Russian Federation), Central NA, Eastern Brazil, and Central SA are regions of high HS for maize yield (Figures 5a and 5c). A high HS effect on soybean yield is found in Central NA and SA, Northern India, Eastern CHN, and the Southwest region of Russia (Figures 5b and 5d).

To quantify crop yield risk level under HS due to projected climate change under two scenarios by the 2090s, we also estimate the coefficient of variation (CV), which is defined as the ratio of standard deviation to mean values of crop yield (Song et al., 2015). Yield risks increase if the variance is larger or the average yield is lower, the CV of crop yield becomes larger (i.e., lower temporal yield stability). Overall, global CV





Figure 5. Maps of the effect of heat stress on maize (a and c) and soybean (b and d) yield (%) for RCP4.5 (a and b) and RCP8.5 (c and d) in the 2090s. Values show [(yield w/ heat stress minus yield w/o heat stress)/(yield w/o heat stress)]*100%.

is higher for soybean than that for maize and higher under RCP8.5 than under RCP4.5. The CV of maize and soybean yield for individual regions is greater in NA, EU, SSEA, and CHN than SA and AF (Table S8 in Supporting Information S1). Furthermore, CV becomes larger when considering HS (E_{Ref}) than without HS ($E_{w/o~HS}$) in the model (Table S8 in Supporting Information S1).

4.2.4. Irrigation

Irrigation enhances maize yield more than that of soybean over the last century and under the two future scenarios (Table S7 in Supporting Information S1 and Figure 4) because of a higher fraction of irrigated area for maize than that for soybean, which is also pointed out by Elliott et al. (2014) and Jägermeyr et al. (2015). The global maize yield with irrigation is estimated to increase by about 5%, whereas 3% for soybean over 1996–2005 (Table S7 in Supporting Information S1), although regional differences exist. Overall, our global results are consistent with previous field measurement studies (Chen et al., 2018; Verma et al., 2005), showing maize yield is more responsive to irrigation than for soybean because of the higher photosynthesis and productivity of maize. On a regional scale, the effects of irrigation over 1996–2005 are most obvious in arid and semiarid areas for both crops, including central and western parts of NA, northeastern CHN, Eastern Australia, Middle East, Central Asia, and western EU (Figure S12a in Supporting Information S1). Also, maize and soybean yields under the irrigated case do not change much in the tropical agriculture regions, that is, SA (Table S7 in Supporting Information S1), because of low irrigated areas and higher precipitation rates. In general, irrigation effect on crop yield becomes less under both scenarios and reduces more in RCP8.5 than RCP4.5. However, the magnitudes do not change much by the 2090s, except for soybean yield in the EU (Table S7 in Supporting Information S1 and Figure 4).

4.2.5. N Input

The effect of N input is stronger for maize than for soybean because soybean is an N-fixing crop (Figure 4, Table S7 and Figure S12 in Supporting Information S1). The stimulation of yield with N input is greater in N-limiting and high N-application regions, including NA, EU, and CHN. These regions are water-limiting regions too. Therefore, irrigation amplifies the N input effect by reducing the water stress effect and enhancing the root carbon that stimulates N uptake (Yang et al., 2009). The interactive N effect increases the yield for both crops in most regions by the 2090s. This is especially under the RCP4.5 scenario with a higher N input amount and improved N availability under favorable environmental conditions (Figure 4). The responses under RCP8.5 are lower because of lower nitrogen input rates per area (Figure S4 in Supporting Information S1) and warmer conditions, weakening the N input effect. The N fertilizer intensification can enhance crop yield in current low-productive regions, such as AF for maize and soybean (Figure 1). The



model results show that agriculture practices in AF result in higher N losses than N added to the soils, which are consistent with other studies (e.g., Lassaletta et al., 2014; Liu et al., 2010; Vitousek et al., 2009). This can be because farmers in AF are considering N management practice, such as 85 percent residue removal at the harvest time (Folberth et al., 2012), which we also simulate in ISAM for the AF region (Text S2 in Supporting Information S1). The model result shows that yields for both crops are increased more under RCP4.5 than the RCP8.5 scenario due to intensified N application rates in the 2090s, especially in AF and SA (Figure 4, Figures S12b and S12c, and Table S7 in Supporting Information S1).

4.2.6. Synergistic Effects on Crop Yield

ISAM results confirm that management factors help offset some of the adverse effects of climate change and limitations of resources (e.g., water and N). However, the interaction between C, water, temperature, and N can be nonlinear; it is expected to produce combined effects than the sum of their individual effects on crop yields. An example here is that the combined effect of N input and irrigation on crop yield is not an additive response of individual forcings. Over the historical time, irrigation and N input offset the decrease in crop yields otherwise caused by drier and N-limitation conditions and increased yield from CO_2 fertilization (Table S7 in Supporting Information S1). When the N and irrigation forcings are combined, synergistic effects enhance crop yields such that the combined effect of N and irrigation on crop yield is more than their individual sum (Figure 4).

Moreover, ISAM results show that crop productivity is colimited by environmental factors, such as $[CO_2]$ and climate. Therefore, management factors under the higher $[CO_2]$ and warmer future climate scenario RCP8.5 may not be able to offset all the crop yield losses by the end of this century. This is the case, for example, for global maize yield w/management case under the RCP8.5 scenario, where yield is estimated to be about 14% lower in the 2090s than in 1996–2005 (Figure 3). These results show that management practices through N and irrigation under socio-economic assumptions of the RCP8.5 scenario and dynamic planting time decisions do not fully compensate for the loss of crop yield from environmental drivers, suggesting the need for more climate change adaptation in the agriculture sector over this century.

4.2.7. Summary and Future Directions

This study aims to quantify the relative importance of two major environmental factors and their interactive effects with management factors and the effect of extreme heat stress on model-estimated productivity of maize and soybean. Our study also provides state-of-the-art estimates of spatial and temporal distributions of crop yield responses to critical high temperature in the reproductive period by identifying heat stress using canopy temperature. As a first step, we show that the model estimated effects of various environmental and management factors on crop yields and their variability in the twentieth century compare well with literature data sets on the global and regional scale.

Accounting for management factors' effects, including those prescribed by the SSPs, model results show that, by 2090s, global maize yield declines by 14% under RCP8.5 and increases under RCP4.5, while soybean yield increases by 20% under RCP4.5 and by 13% under RCP8.5. Yet Kriegler et al. (2017) find that global cereal crop yield would need to increase by 60% between 2005 and 2100 under both scenarios to fulfill future food demand. These results suggest that the management practices considered in this study for maize and soybean will not be sufficient to satisfy the future food demand, implying that additional advanced technologies and management practices will be needed to intensify crop productivity in the future.

Looking beyond this study, we plan to improve the treatment of management practices so that observed yield and future trend projections can be improved (cf., Alexandratos & Bruinsma, 2012). For instance, ISAM does not consider different irrigation sources (such as irrigated water from groundwater pumping) and pathways (flood irrigation, drip irrigation, and sprinkler irrigation), which may affect hydrological cycles (Jägermeyr et al., 2015; Leng et al., 2017). Thus, improving water use processes for each crop type in the model can improve the estimation of global human water usage, crop yield, and irrigation demand for crops (Webber et al., 2016). Additionally, it should be noted that the input data such as climate and N inputs are primary but inevitable sources of uncertainties in the estimated future crop yields. In the future analysis, we plan to study the crop yields using the ensemble of multimodel results rather than single model results. We also plan to correct model simulated precipitation fields, including variability and other higher moments, using different bias correction approaches (e.g., Cannon, 2016; Li et al., 2010). Although we use



state-of-the-art N input data under SSPs from Coupled Model Intercomparison Project Phase 6 (CMIP6) (Lawrence et al., 2016), future studies should consider different N application methods, forms, timing, and levels. These advancements and additional model analyses with other crops are needed to mitigate the effects of climate change and extremes on the agriculture system.

Conflicts of Interest

The authors declare no conflicts of interest relevant to this study.

Data Availability Statement

We provide the following data sets: (a) Spatial distribution of mean maize and soybean production (unit: tonnes/year for a 0.5° x 0.5° grid) from ISAM simulations and literature averaged for the time period 1996–2005 (Figure 2). (b) Spatial distribution of irrigated and rainfed areas for maize and soybean (unit: fraction of grid cell) for 1990–2015. (c) Spatial distribution of N fertilizer (unit: kgN/ha for 0.5° x 0.5° grid) applied over the maize and soybean areas for 1990–2100. The data for 2016–2100 are based on scenarios. (d) ISAM simulated the spatial distribution of maize and soybean yields (unit: t/ha) based on various model experiments described in Table 1. All the data are stored in NetCDF files. The data can be accessed from the ISAM website: http://climate.atmos.uiuc.edu/Lin_Cropyields/.

References

- Ainsworth, E. A. (2008). Rice production in a changing climate: A meta-analysis of responses to elevated carbon dioxide and elevated ozone concentration. *Global Change Biology*. https://doi.org/10.1111/j.1365-2486.2008.01594.x
- Ainsworth, E. A., Davey, P. A., Bernacchi, C. J., Dermody, O. C., Heaton, E. A., Moore, D. J., et al. (2002). A meta-analysis of elevated (CO₂) effects on soybean (Glycine max) physiology, growth and yield. *Global Change Biology*, *8*(8), 695–709. https://doi. org/10.1046/j.1365-2486.2002.00498.x
- Alemayehu, Y., Shewarega, M., Resources, N., & Resources, N. (2015). Growth and yield responses of Maize (Zea Mays L.) to different nitrogen rates under rain-fed condition in Dilla area, Southern Ethiopia, *Journal of Natural Sciences Research*, *5*(23), 40–46.
- Alexandratos, N., & Bruinsma, J. (2012). World agriculture towards 2030/2050: The 2012 revision, ESA Working paper No. 12-03. Rome. FAO.
- Asseng, S., Jamieson, P. D., Kimball, B., Pinter, P., Sayre, K., Bowden, J. W., & Howden, S. M. (2004). Simulated wheat growth affected by rising temperature, increased water deficit and elevated atmospheric CO₂. *Field Crops Research*. https://doi.org/10.1016/ S0378-4290(03)00154-0
- Barman, R., Jain, A. K., & Liang, M. (2014a). Climate-driven uncertainties in modeling terrestrial energy and water fluxes: A site-level to global-scale analysis. *Global Change Biology*. https://doi.org/10.1111/gcb.12473
- Barman, R., Jain, A. K., & Liang, M. (2014b). Climate-driven uncertainties in modeling terrestrial gross primary production: A site level to global-scale analysis. *Global Change Biology*. https://doi.org/10.1111/gcb.12474
- Bernacchi, C. J., Leakey, A. D. B., Heady, L. E., Morgan, P. B., Dohleman, F. G., McGrath, J. M., et al. (2006). Hourly and seasonal variation in photosynthesis and stomatal conductance of soybean grown at future CO₂ and ozone concentrations for 3 years under fully open-air field conditions. *Plant, Cell and Environment*, 29(11), 2077–2090. https://doi.org/10.1111/j.1365-3040.2006.01581.x
- Bolanos, J., & Edmeades, G. O. (1996). The importance of the anthesis-silking interval in breeding for drought tolerance in tropical maize. *Field Crops Research*, 48, 65–80.
- Cafaro La Menza, N., Monzon, J. P., Specht, J. E., & Grassini, P. (2017). Is soybean yield limited by nitrogen supply? Field Crops Research, 213, 204–212. https://doi.org/10.1016/j.fcr.2017.08.009
- Cannon, A. F. (2016). Multivariate bias correction of climate model output: Matching marginal distributions and intervariable dependence structure. *Journal of Climate*, 29, 7045–7064. https://doi.org/10.1175/JCLI-D-15-0679.1
- Challinor, A. J., Watson, J., Lobell, D. B., Howden, S. M., Smith, D. R., & Chhetri, N. (2014). A meta-analysis of crop yield under climate change and adaptation. *Nature Climate Change*, 4(4), 287–291. https://doi.org/10.1038/nclimate2153
- Chen, F., Xu, X., Barlage, M., Rasmussen, R., Shen, S., Miao, S., & Zhou, G. (2018). Memory of irrigation effects on hydroclimate and its modeling challenge. *Environmental Research Letters*, 13(6). https://doi.org/10.1088/1748-9326/aab9df
- Dangal, S. R. S., Tian, H., Xu, R., Chang, J., Canadell, J. G., Ciais, P., et al. (2019). Global nitrous oxide emissions from pasturelands and rangelands: Magnitude, spatiotemporal patterns, and attribution. *Global Biogeochemical Cycles*, *33*(2), 200–222. https://doi. org/10.1029/2018gb006091
- Deryng, D., Conway, D., Ramankutty, N., Price, J., & Warren, R. (2014). Global crop yield response to extreme heat stress under multiple climate change futures. *Environmental Research Letters*, 9(3). https://doi.org/10.1088/1748-9326/9/3/034011
- Deryng, D., Elliott, J., Folberth, C., Müller, C., Pugh, T. A. M., Boote, K. J., et al. (2016). Regional disparities in the beneficial effects of rising CO₂ concentrations on crop water productivity. *Nature Climate Change*, 6(8), 786–790. https://doi.org/10.1038/nclimate2995
- El Masri, B., Shu, S., & Jain, A. K. (2015). Implementation of a dynamic rooting depth and phenology into a land surface model: Evaluation of carbon, water, and energy fluxes in the high latitude ecosystems. Agricultural and Forest Meteorology. https://doi.org/10.1016/j. agrformet.2015.06.002
- Elliott, J., Deryng, D., Müller, C., Frieler, K., Konzmann, M., Gerten, D., et al. (2014). Constraints and potentials of future irrigation water availability on agricultural production under climate change. *Proceedings of the National Academy of Sciences of the United States of America*, 111, 3239–3244. https://doi.org/10.1073/pnas.1222474110

Acknowledgments

This work is supported by the US National Science Foundation (NSF-AGS-12-43071, NSF- 831361857). The authors would also like to acknowledge high-performance computing support from Cheyenne (doi:10.5065/D6RX-99HX) provided by NCAR's Computational and Information Systems Laboratory, sponsored by the National Science Foundation.



- Elliott, J., Müller, C., Deryng, D., Chryssanthacopoulos, J., Boote, K. J., Büchner, M., et al. (2015). The global gridded crop model intercomparison: Data and modeling protocols for phase 1 (v1.0). *Geoscientific Model Development*, 8(2), 261–277. https://doi.org/10.5194/ gmd-8-261-2015
- El-Masri, B., Barman, R., Meiyappan, P., Song, Y., Liang, M., & Jain, A. K. (2013). Carbon dynamics in the Amazonian Basin: Integration of eddy covariance and ecophysiological data with a land surface model. *Agricultural and Forest Meteorology*. https://doi.org/10.1016/j. agrformet.2013.03.011

FAOstat. (2017). Retrieved from http://faostat3.fao.org/home/E

- Fodor, N., Challinor, A., Droutsas, I., Ramirez-Villegas, J., Zabel, F., Koehler, A. K., & Foyer, C. H. (2017). Integrating plant science and crop modeling: Assessment of the impact of climate change on soybean and maize production. *Plant and Cell Physiology*, 58(11), 1833–1847. https://doi.org/10.1093/pcp/pcx141
- Folberth, C., Gaiser, T., Abbaspour, K. C., Schulin, R., & Yang, H. (2012). Regionalization of a large-scale crop growth model for sub-Saharan Africa: Model setup, evaluation, and estimation of maize yields. *Agriculture, Ecosystems & Environment, 151*, 21–33. https://doi. org/10.1016/j.agee.2012.01.026
- Gabaldón-Leal, C., Webber, H., Otegui, M. E., Slafer, G. A., Ordóñez, R. A., Gaiser, T., et al. (2016). Modelling the impact of heat stress on maize yield formation. *Field Crops Research*. https://doi.org/10.1016/j.fcr.2016.08.013
- Gahlot, S., Lin, T., Jain, A. K., Roy, S. B., Sehgal, V. K., & Dhakar, R. (2020). Impact of environmental changes and land management practices on wheat production in India. *Earth System Dynamics*, 11(3), 641–652. https://doi.org/10.5194/esd-11-641-2020
- Gahlot, S., Shu, S., Jain, A. K., & Baidya Roy, S. (2017). Estimating trends and variation of net biome productivity in India for 1980–2012 using a land surface model. *Geophysical Research Letters*. https://doi.org/10.1002/2017GL075777
- Gehl, R. J., Schmidt, J. P., Maddux, L. D., & Gordon, W. B. (2005). Corn yield response to nitrogen rate and timing in sandy irrigated soils. Agronomy Journal, 97(4), 1230–1238. https://doi.org/10.2134/agronj2004.0303
- Getachew, M., & Belete, K. (2013). Yield related traits and yield of maize (Zea Mays L.) as affected by green manuring and nitrogen levels at Mizan Teferi, South-west. International Journal of Agronomy and Plant Production, 4(7), 1462–1473.
- Gourdji, S. M., Sibley, A. M., & Lobell, D. B. (2013). Global crop exposure to critical high temperatures in the reproductive period: Historical trends and future projections. *Environmental Research Letters*, 8(2). https://doi.org/10.1088/1748-9326/8/2/024041
- Gray, S. B., Dermody, O., Klein, S. P., Locke, A. M., McGrath, J. M., Paul, R. E., et al. (2016). Intensifying drought eliminates the expected benefits of elevated carbon dioxide for soybean. *Nature Plants*, 2, 1–26. https://doi.org/10.1038/nplants.2016.132
- Hammad, H. M., Ahmad, A., Wajid, A., & Akhter, J. (2011). Maize response to time and rate of nitrogen application. Pakistan Journal of Botany, 43(4), 1935–1942.
- Harris, I., Jones, P. D., Osborn, T. J., & Lister, D. H. (2014). Updated high-resolution grids of monthly climatic observations The CRU TS3.10 dataset. International Journal of Climatology. https://doi.org/10.1002/joc.3711
- Hurtt, G. C., Chini, L., Sahajpal, R., Frolking, S., Bodirsky, B. L., Calvin, K., et al. (2020). Harmonization of global land use change and management for the period 850–2100 (LUH2) for CMIP6. *Geoscientific Model Development*, 13, 5425–5464. https://doi.org/10.5194/ gmd-13-5425-2020
- Iizumi, T., Furuya, J., Shen, Z., Kim, W., Okada, M., Fujimori, S., et al. (2017). Responses of crop yield growth to global temperature and socio-economic changes. *Scientific Reports*. https://doi.org/10.1038/s41598-017-08214-4
- Iizumi, T., Yokozawa, M., Sakurai, G., Travasso, M. I., Romanernkov, V., Oettli, P., et al. (2014). Historical changes in global yields: Major cereal and legume crops from 1982 to 2006. *Global Ecology and Biogeography*, 23(3), 346–357. https://doi.org/10.1111/geb.12120
- Jägermeyr, J., Gerten, D., Heinke, J., Schaphoff, S., Kummu, M., & Lucht, W. (2015). Water savings potentials of irrigation systems: Global simulation of processes and linkages. *Hydrology and Earth System Sciences*, 19(7), 3073–3091. https://doi.org/10.5194/hess-19-3073-2015
- Jagtap, S. S. M., & Jones, J. W. (2002). Adaptation and evaluation of the CROPGRO-soybean model to predict regional yield and production. *Agriculture, Ecosystems & Environment*, 93(1–3), 73–85. https://doi.org/10.1016/s0167-8809(01)00358-9
- Jain, A., Yang, X., Kheshgi, H., McGuire, A. D., Post, W., & Kicklighter, D. (2009). Nitrogen attenuation of terrestrial carbon cycle response to global environmental factors. *Global Biogeochemical Cycles*, 23(4), GB4028. https://doi.org/10.1029/2009GB003519
- Jain, A. K., Meiyappan, P., Song, Y., & House, J. I. (2013). CO₂ emissions from land-use change affected more by nitrogen cycle, than by the choice of land-cover data. *Global Change Biology*, 19(9), 2893–2906. https://doi.org/10.1111/gcb.12207
- Jain, A. K., Tao, Z., Yang, X., & Gillespie, C. (2006). Estimates of global biomass burning emissions for reactive greenhouse gases (CO, NMHCs, and NOx) and CO₂. Journal of Geophysical Research. https://doi.org/10.1029/2005JD006237
- Kheshgi, H. S., Jain, A. K., & Wuebbles, D. J. (1999). Model-based estimation of the global carbon budget and its uncertainty from carbon dioxide and carbon isotope records. *Journal of Geophysical Research*, 104(D24), 31127–31143. https://doi.org/10.1029/1999JD900992
- Kim, S., Sicher, R., Bae, H., Gitz, D., Baker, J., Timlin, D., et al. (2006). Canopy photosynthesis, evapotranspiration, leaf nitrogen, and transcription profiles of maize in response to CO₂ enrichment. *Global Change Biology*, 12(3), 588–600. https://doi. org/10.1111/j.1365-2486.2006.01110.x
- Kriegler, E., Bauer, N., Popp, A., Hupenöder, F., Leimbach, M., Strefler, J., et al. (2017). Fossil-fueled development (SSP5): An energy and resource intensive scenario for the 21st century. *Global Environmental Change*, 42, 297–315. https://doi.org/10.1016/j.gloenvcha.2016.05.015
- Ladha, J. K., Tirol-Padre, A., Reddy, C. K., Cassman, K. G., Verma, S., Powlson, D. S., et al. (2016). Global nitrogen budgets in cereals: A 50-year assessment for maize, rice and wheat production systems. *Scientific Reports*, 6(1). https://doi.org/10.1038/srep19355
- Lamarque, J.-F., Kyle, G. P., Meinshausen, M., Riahi, K., Smith, S. J., Vuuren, D. P. V., et al. (2011). Global and regional evolution of shortlived radiatively-active gases and aerosols in the Representative Concentration Pathways. *Climatic Change*, 109(1–2), 191–212. https:// doi.org/10.1007/s10584-011-0155-0
- Lassaletta, L., Billen, G., Grizzetti, B., Anglade, J., & Garnier, J. (2014). 50 year trends in nitrogen use efficiency of world cropping systems: The relationship between yield and nitrogen input to cropland. *Environmental Research Letters*, 9(10). https://doi.org/10.1088/1748-9326/9/10/105011
- Lawrence, D. M., Hurtt, G. C., Arneth, A., Brovkin, V., Calvin, K. V., Jones, A. D., et al. (2016). The land use model intercomparison project (LUMIP) contribution TO CMIP6: Rationale and experimental design. *Geoscientific Model Development*, 9(9), 2973–2998. https://doi.org/10.5194/gmd-9-2973-2016
- Le Quéré, C., Andrew, R. M., Friedlingstein, P., Sitch, S., Hauck, J., Pongratz, J., et al. (2018). Global carbon budget 2018. Earth System Science Data, 10, 2141–2194. https://doi.org/10.5194/essd-10-2141-2018
- Leakey, A. D. B., Uribelarreà, M., Ainsworth, E. A., Naidu, S. L., Rogers, A., Ort, D. R., & Long, S. P. (2006). Photosynthesis, productivity, and yield of maize are not affected by open-air elevation of CO₂ concentration in the absence of drought. *Plant Physiology*. https://doi.org/10.1104/pp.105.073957



- Leng, G., Leung, L. R., & Huang, M. (2017). Significant impacts of irrigation water sources and methods on modeling irrigation effects in the ACME Land Model. *Journal of Advances in Modeling Earth Systems*. https://doi.org/10.1002/2016MS000885
- Levis, S., Badger, A., Drewniak, B., Nevison, C., & Ren, X. (2018). CLMcrop yields and water requirements: Avoided impacts by choosing RCP 4.5 over 8.5. *Climatic Change*, 146(3–4), 501–515. https://doi.org/10.1007/s10584-016-1654-9
- Li, H., Sheffield, J., & Wood, E. F. (2010). Bias correction of monthly precipitation and temperature fields from intergovernmental panel on climate change AR4 models using equidistant quantile matching. *Journal of Geophysical Research*, *115*, D10101. https://doi. org/10.1029/2009JD012882
- Lin, T. S., Gahlot, S., Shu, S., Jain, A. K., & Kheshgi, H. S. (2017). Studying the impacts of environmental factors and agricultural management on methane emissions from rice paddies using a land surface model. AGU Fall Meeting.
- Liu, J., You, L., Amini, M., Obersteiner, M., Herrero, M., Zehnder, A. J. B., & Yang, H. (2010). A high-resolution assessment on global nitrogen flows in cropland. Proceedings of the National Academy of Sciences of the United States of America. https://doi.org/10.1073/ pnas.0913658107
- Lobell, D. B., Hicke, J. A., Asner, G. P., Field, C. B., Tucker, C. J., & Los, S. O. (2002). Satellite estimates of productivity and light use efficiency in United States agriculture 1982–98. Global Change Biology, 8(8), 722–735. https://doi.org/10.1046/j.1365-2486.2002.00503.x
- Lobell, D. B., Roberts, M. J., Schlenker, W., Braun, N., Little, B. B., Rejesus, R. M., & Hammer, G. L. (2014). Greater sensitivity to drought accompanies maize yield increase in the U.S. midwest. Science, 344(6183), 516–519. https://doi.org/10.1126/science.1251423
- Lombardozzi, D. L., Bonan, G. B., Levis, S., & Lawrence, D. M. (2018). Changes in wood biomass and crop yields in response to projected CO₂, O₃, Nitrogen deposition, and climate. *Journal of Geophysical Research: Biogeosciences*, 123(10), 3262–3282. https://doi. org/10.1029/2018jg004680
- McGrath, J. M., & Lobell, D. B. (2013). Regional disparities in the CO₂ fertilization effect and implications for crop yields. *Environmental Research Letter*. https://doi.org/10.1088/1748-9326/8/1/014054
- Meinshausen, M., Smith, S. J., Calvin, K., Daniel, J. S., Kainuma, M. L. T., Lamarque, J., et al. (2011). The RCP greenhouse gas concentrations and their extensions from 1765 to 2300. Climatic Change. https://doi.org/10.1007/s10584-011-0156-z
- Monfreda, C., Ramankutty, N., & Foley, J. A. (2008). Farming the planet: 2. Geographic distribution of crop areas, yields, physiological types, and net primary production in the year 2000. Global Biogeochemical Cycles, 22(1), 1–19. https://doi.org/10.1029/2007GB002947
- Mueller, N. D., Gerber, J. S., Johnston, M., Ray, D. K., Ramankutty, N., & Foley, J. A. (2012). Closing yield gaps through nutrient and water management. Nature. https://doi.org/10.1038/nature11420
- Müller, C., Elliott, J., Chryssanthacopoulos, J., Arneth, A., Balkovic, J., Ciais, P., et al. (2017). Global gridded crop model evaluation: Benchmarking, skills, deficiencies and implications. *Geoscientific Model Development*. https://doi.org/10.5194/gmd-10-1403-2017
- Müller, C., Elliott, J., Chryssanthacopoulos, J., Deryng, D., Folberth, C., Pugh, T. A. M., & Schmid, E. (2015). Implications of climate mitigation for future agricultural production. *Environmental Research Letters*. https://doi.org/10.1088/1748-9326/10/12/125004
- Müller, C., Elliott, J., Kelly, D., Arneth, A., Balkovic, J., Ciais, P., et al. (2019). The global gridded crop model Intercomparison phase 1 simulation dataset (pp. 1–22). https://doi.org/10.1038/s41597-019-0023-8
- Nishina, K., Ito, A., Hanasaki, N., & Hayashi, S. (2017). Reconstruction of spatially detailed global map of NH_4^+ and NO_3^- application in synthetic nitrogen fertilizer. Earth System Science Data.
- Niyogi, D., Liu, X., Andresen, J., Song, Y., Jain, A. K., Kellner, O., et al. (2015). Crop models capture the impacts of climate variability on corn yield. *Geophysical*, 1, 3356–3363. https://doi.org/10.1002/2015GL063841.We
- O'Neill, B. C., Kriegler, E., Riahi, K., Ebi, K. L., Hallegatte, S., Carter, T. R., et al. (2014). A new scenario framework for climate change research: The concept of shared socio-economic pathways. *Climatic Change*. https://doi.org/10.1007/s10584-013-0905-2
- Pandey, R. K., Maranville, J. W., & Admou, A. (2000). Deficit irrigation and nitrogen effects on maize in a sahelian environment. Agricultural Water Management, 46(1), 1–13. https://doi.org/10.1016/s0378-3774(00)00073-1
- Popp, A., Calvin, K., Fujimori, S., Havlik, P., Humpenöder, F., Stehfest, E., et al. (2017). Land-use futures in the shared socio-economic pathways. *Global Environmental Change*, 42, 331–345. https://doi.org/10.1016/j.gloenvcha.2016.10.002
- Portmann, F. T., Siebert, S., & Döll, P. (2010). MIRCA2000-Global monthly irrigated and rainfed crop areas around the year 2000: A new high-resolution data set for agricultural and hydrological modeling. *Global Biogeochemical Cycles*. https://doi.org/10.1029/2008gb003435
- Porwollik, V., Müller, C., Elliott, J., Chryssanthacopoulos, J., Iizumi, T., Ray, D. K., et al. (2017). Spatial and temporal uncertainty of crop yield aggregations. European Journal of Agronomy, 88, 10–21. https://doi.org/10.1016/j.eja.2016.08.006
- Quéré, C., Andrew, R., Friedlingstein, P., Sitch, S., Hauck, J., Pongratz, J., et al. (2018). Global carbon budget 2018. Earth System Science Data. https://doi.org/10.5194/essd-10-2141-2018
- Rahimi-Moghaddam, S., Kambouzia, J., & Deihimfard, R. (2018). Adaptation strategies to lessen negative impact of climate change on grain maize under hot climatic conditions: A model-based assessment. Agricultural and Forest Meteorology, 253–254, 1–14. https://doi. org/10.1016/j.agrformet.2018.01.032
- Ren, X., Weitzel, M., O'Neill, B. C., Lawrence, P., Meiyappan, P., Levis, S., et al. (2018). Avoided economic impacts of climate change on agriculture: Integrating a land surface model (CLM) with a global economic model (iPETS). *Climatic Change*, 146(3–4), 517–531. https:// doi.org/10.1007/s10584-016-1791-1
- Rosenzweig, C., Elliott, J., Deryng, D., Ruane, A. C., Müller, C., Arneth, A., et al. (2014). Assessing agricultural risks of climate change in the 21st century in a global gridded crop model intercomparison. *Proceedings of the National Academy of Sciences*, 111(9), 3268–3273. https://doi.org/10.1073/pnas.1222463110
- Ruiz-Vera, U. M., Siebers, M., Gray, S. B., Drag, D. W., Rosenthal, D. M., Kimball, B. A., et al. (2013). Global warming can negate the expected CO₂ stimulation in photosynthesis and productivity for soybean grown in the midwestern United States. *Plant Physiology*, 162(1), 410–423. https://doi.org/10.1104/pp.112.211938
- Ruiz-Vera, U. M., Siebers, M. H., Drag, D. W., Ort, D. R., & Bernacchi, C. J. (2015). Canopy warming caused photosynthetic acclimation and reduced seed yield in maize grown at ambient and elevated (CO₂). *Global Change Biology*, 21(11), 4237–4249. https://doi.org/10.1111/ gcb.13013
- Rustad, L. E., Campbell, J. L., Marion, G. M., Norby, R. J., Mitchell, M. J., Hartley, A. E., et al. (2001). A meta-analysis of the response of soil respiration, net nitrogen mineralization, and aboveground plant growth to experimental ecosystem warming. *Oecologia*. https:// doi.org/10.1007/s004420000544
- Sage, T. L., Bagha, S., Lundsgaard-Nielsen, V., Branch, H. A., Sultmanis, S., & Sage, R. F. (2015). The effect of high temperature stress on male and female reproduction in plants. *Field Crops Research*, 182, 30–42. https://doi.org/10.1016/j.fcr.2015.06.011
- Sakurai, G., Iizumi, T., Nishimori, M., & Yokozawa, M. (2014). How much has the increase in atmospheric CO2 directly affected past (pp. 1–5). https://doi.org/10.1038/srep04978

- Salvagiotti, F., Cassman, K. G., Specht, J. E., Walters, D. T., Weiss, A., & Dobermann, A. (2008). Nitrogen uptake, fixation and response to fertilizer N in soybeans: A review. *Field Crops Research*, *108*(1), 1–13. https://doi.org/10.1016/j.fcr.2008.03.001
- Siebert, S., Ewert, F., Eyshi Rezaei, E., Kage, H., & Gra
 ß, R. (2014). Impact of heat stress on crop yield On the importance of considering canopy temperature. *Environmental Research Letters*, 9(4). https://doi.org/10.1088/1748-9326/9/4/044012
- Song, Y., Cervarich, M., Jain, A. K., Kheshgi, H. S., Landuyt, W., & Cai, X. (2016). The interplay between bioenergy grass production and water resources in the United States of America. *Environmental Science & Technology*, 50(6), 3010–3019. https://doi.org/10.1021/acs. est.5b05239
- Song, Y., Jain, A. K., Landuyt, W., Kheshgi, H. S., & Khanna, M. (2015). Estimates of biomass yield for perennial bioenergy grasses in the USA. *Bioenergy Research*, 8(2), 688–715. https://doi.org/10.1007/s12155-014-9546-1
- Song, Y., Jain, A. K., & McIsaac, G. F. (2013). Implementation of dynamic crop growth processes into a land surface model: Evaluation of energy, water and carbon fluxes under corn and soybean rotation. *Biogeosciences*, 10(12), 8039–8066. https://doi.org/10.5194/ bg-10-8039-2013
- Teixeira, E. I., Fischer, G., Van Velthuizen, H., Walter, C., & Ewert, F. (2013). Global hot-spots of heat stress on agricultural crops due to climate change. Agricultural and Forest Meteorology, 170, 206–215. https://doi.org/10.1016/j.agrformet.2011.09.002
- Tian, H., Yang, J., Lu, C., Xu, R., Canadell, J. G., Jackson, R. B., et al. (2018). The global N₂O model intercomparison project. Bulletin of the American Meteorological Society. https://doi.org/10.1175/BAMS-D-17-0212.1
- van Vuuren, D. P., Edmonds, J., Kainuma, M., Riahi, K., Thomson, A., Hibbard, K., et al. (2011). The representative concentration pathways: An overview. *Climatic Change*. https://doi.org/10.1007/s10584-011-0148-z
- Verma, S. B., Dobermann, A., Cassman, K. G., Walters, D. T., Knops, J. M., Arkebauer, T. J., et al. (2005). Annual carbon dioxide exchange in irrigated and rainfed maize-based agroecosystems. *Agricultural and Forest Meteorology*, 131(1–2), 77–96. https://doi.org/10.1016/j. agrformet.2005.05.003
- Viovy, N. (2016). Retrieved from CRUNCEP data set. ftp://nacp.ornl.gov/synthesis/2009/frescati/temp/land_use_change/original/readme. htm
- Vitousek, P. M., Naylor, R., Crews, T., David, M. B., Drinkwater, L. E., Holland, E., et al. (2009). Nutrient imbalances in agricultural development. Science. https://doi.org/10.1126/science.1170261
- Webber, H., Gaiser, T., Oomen, R., Teixeira, E., Zhao, G., Wallach, D., et al. (2016). Uncertainty in future irrigation water demand and risk of crop failure for maize in Europe. *Environmental Research Letters*. https://doi.org/10.1088/1748-9326/11/7/074007
- Webber, H., Martre, P., Asseng, S., Kimball, B., White, J., Ottman, M., et al. (2017). Canopy temperature for simulation of heat stress in irrigated wheat in a semi-arid environment: A multi-model comparison. *Field Crops Research*. https://doi.org/10.1016/j.fcr.2015.10.009Wirsenius, S. (2000). *Human use of land and organic materials*. Thesis for the degree of Doctor of Philosophy (pp. 255). Göteborg, Sweden.
- Department of Physical Resource Theory, Chalmers University of Technology and Göteborg University. Yamaguchi, J. (1978). Respiration and growth efficiency in relation to crop productivity. *Journal of the Faculty of Agriculture*. Hokkuido
- University, 59(1), 59–129. Yang, X., Wittig, V., Jain, A. K., & Post, W. (2009). Integration of nitrogen cycle dynamics into the integrated science assessment
- Yang, X., Wittig, V., Jain, A. K., & Post, W. (2009). Integration of nitrogen cycle dynamics into the integrated science assessment model for the study of terrestrial ecosystem responses to global change. *Global Biogeochemical Cycles*, 23(4), 1–18. https://doi. org/10.1029/2009GB003474
- You, L., Wood, S., Wood-Sichra, U., & Wu, W. (2014). Generating global crop distribution maps: From census to grid. *Agricultural Systems*, 127, 53–60. https://doi.org/10.1016/j.agsy.2014.01.002
- Zhang, B., Tian, H., Lu, C., Dangal, S. R. S., Yang, J., & Pan, S. (2017). Global manure nitrogen production and application in cropland during 1860–2014: A 5 arcmin gridded global dataset for Earth system modeling. *Earth System Science Data*, 9(2), 667–678. https://doi. org/10.5194/essd-9-667-2017
- Zhang, Y., Yu, G., Yang, J., Wimberly, M. C., Zhang, X., Tao, J., et al. (2013). Climate-driven global changes in carbon use efficiency. Global Ecology and Biogeography, 23(2), 144–155. https://doi.org/10.1111/geb.12086

References From the Supporting Information

- Akinnifesi, F. K., Makumba, W., Sileshi, G., Ajayi, O. C., & Mweta, D. (2007). Synergistic effect of inorganic N and P fertilizers and organic inputs from Gliricidia sepium on productivity of intercropped maize in Southern Malawi. *Plant and Soil*, 294(1–2), 203–217. https:// doi.org/10.1007/s11104-007-9247-z
- Al-Ithawi, B., Deibert, E. J., & Olson, R. A. (1980). Applied N and moisture level effects on yield, depth of root activity, and nutrient uptake by soybeans1. Agronomy Journal, 72(5), 827. https://doi.org/10.2134/agronj1980.00021962007200050031x
- Arora, V. K. (2003). Simulating energy and carbon fluxes over winter wheat using coupled land surface and terrestrial ecosystem models. Agricultural and Forest Meteorology, 118(1-2), 21-47. https://doi.org/10.1016/S0168-1923(03)00073-X
- Attia, A., Shapiro, C., Kranz, W., Mamo, M., & Mainz, M. (2015). Improved yield and nitrogen use efficiency of corn following soybean in irrigated sandy loams. *Soil Science Society of America Journal*, 79(6), 1693. https://doi.org/10.2136/sssaj2015.05.0200
- Bonan, G. B., Lawrence, P. J., Oleson, K. W., Levis, S., Jung, M., Reichstein, M., et al. (2011). Improving canopy processes in the Community Land Model version 4 (CLM4) using global flux fields empirically inferred from. *FLUXNET data*, 116, 1–22. https://doi. org/10.1029/2010JG001593
- Boyer, C. N., Larson, J. A., Roberts, R. K., McClure, A. T., Tyler, D. D., & Zhou, V. (2013). Stochastic corn yield response functions to nitrogen for corn after corn, corn after cotton, and corn after soybeans. *Journal of Agricultural & Applied Economics*, 45(4), 669–681. https:// doi.org/10.1017/s1074070800005198
- De Bruin, J. L., & Pedersen, P. (2008). Effect of row spacing and seeding rate on soybean yield. Agronomy Journal. https://doi.org/10.2134/ agronj2007.0106
- Challinor, A., Wheeler, T., Craufurd, P., & Slingo, J. (2005). Simulation of the impact of high temperature stress on annual crop yields. *Agricultural and Forest Meteorology*, 135(1–4), 180–189. https://doi.org/10.1016/j.agrformet.2005.11.015
- Chen, H., Dickinson, R. E., Dai, Y., & Zhou, L. (2011). Sensitivity of simulated terrestrial carbon assimilation and canopy transpiration to different stomatal conductance and carbon assimilation schemes. *Climate Dynamics*. https://doi.org/10.1007/s00382-010-0741-2
- Déqué, M., Rowell, D. P., Lüthi, D., Giorgi, F., Christensen, J. H., Rockel, B., et al. (2007). An intercomparison of regional climate simulations for Europe: Assessing uncertainties in model projections. *Climatic Change*. https://doi.org/10.1007/s10584-006-9228-x
- Derby, N. E., Steele, D. D., Terpstra, J., Knighton, R. E., & Casey, F. X. M. (2005). Interactions of nitrogen, weather, soil, and irrigation on corn yield. Agronomy Journal, 97(5), 1342–1351. https://doi.org/10.2134/agronj2005.0051



- Deryng, D., Sacks, W. J., Barford, C. C., & Ramankutty, N. (2011). Simulating the effects of climate and agricultural management practices on global crop yield. *Global Biogeochemical Cycles*, 25(2). https://doi.org/10.1029/2009GB003765
- Doraiswamy, P. C., McCarty, G. W., Hunt, E. R., Yost, R. S., Doumbia, M., & Franzluebbers, A. J. (2007). Modeling soil carbon sequestration in agricultural lands of Mali. Agricultural Systems. https://doi.org/10.1016/j.agsy.2005.09.011
- Doran, J. W., Wilhelm, W. W., & Power, J. F. (1984). Crop residue removal and soil productivity with no-till corn, sorghum, and soybean. Soil Science Society of America Journal, 48, 640–645.
- Dugje, I. Y., Omoigui, L. O., Ekeleme, F., Bandyopadhyay, R., Lava Kumar, P., & Kamara, A. Y. (2009). Farmers' guide to soybean production in northern Nigeria (pp. 21).
- Hawkins, E., Osborne, T. M., Ho, C. K., & Challinor, A. J. (2013). Calibration and bias correction of climate projections for crop modelling: An idealised case study over Europe. Agricultural and Forest Meteorology, 170, 19–31. https://doi.org/10.1016/j.agrformet.2012.04.007
- Ho, C. K., Stephenson, D. B., Collins, M., Ferro, C. A. T., & Brown, S. J. (2012). Calibration strategies a source of additional uncertainty in climate change projections. *Bulletin of the American Meteorological Society*, 93(1), 21–26. https://doi.org/10.1175/2011BAMS3110.1
- Krausmann, F., Erb, K., Gingrich, S., Lauk, C., & Haberl, H. (2008). Global patterns of socio-economic biomass flows in the year 2000: A comprehensive assessment of supply, consumption and constraints. *Ecological Economics*, 65(3), 471–487. https://doi.org/10.1016/j. ecolecon.2007.07.012
- Kucharik, C. J., & Brye, K. R. (2003). Integrated biosphere simulator (IBIS) yield and nitrate loss predictions for Wisconsin maize receiving varied amounts of nitrogen fertilizer. Journal of Environmental Quality, 32(1), 247–268.
- Leng, G., Huang, M., Tang, Q., & Leung, L. R. (2015). A modeling study of irrigation effects on global surface water and groundwater resources under a changing climate. Journal of Advances in Modeling Earth Systems. https://doi.org/10.1002/2015MS000437
- Lombardozzi, D. L., Lu, Y., Lawrence, P. J., Lawrence, D. M., Swenson, S., Oleson, K. W., et al. (2020). Simulating agriculture in the community land model version 5. Journal of Geophysical Research: Biogeosciences, 125(8). https://doi.org/10.1029/2019jg005529
- Manderscheid, R., Erbs, M., & Weigel, H. (2014). Interactive effects of free-air CO₂ enrichment and drought stress on maize growth. *European Journal of Agronomy*, 52, 11–21. https://doi.org/10.1016/j.eja.2011.12.007
- Meehl, G. A., Hu, A., Tebaldi, C., Arblaster, J. M., Washington, W. M., Teng, H., et al. (2012). Relative outcomes of climate change mitigation related to global temperature versus sea-level rise. *Nature Climate Change*. https://doi.org/10.1038/nclimate1529
- Morgan, P. B., Bollero, G. A., Nelson, R. L., Dohleman, F. G., & Long, S. P. (2005). Smaller than predicted increase in aboveground net primary production and yield of field-grown soybean under fully open-air (CO₂) elevation. *Global Change Biology*, 11(10), 1856–1865. https://doi.org/10.1111/j.1365-2486.2005.001017.x
- Moriondo, M., Giannakopoulos, C., & Bindi, M. (2010). Climate change impact assessment: The role of climate extremes in crop yield simulation. *Climatic Change*, 104(3–4), 679–701. https://doi.org/10.1007/s10584-010-9871-0
- Nangia, V., Turral, H., & Molden, D. (2008). Increasing water productivity with improved N fertilizer management. Irrigation and Drainage Systems, 22(3–4), 193–207. https://doi.org/10.1007/s10795-008-9051-9
- Oleson, K. W., Lawrence, D. M., Bonan, G. B., Drewniak, B., Huang, M., Charles, D., et al. (2013). NCAR/TN-503+STR NCAR Technical note, (July)
- Puntel, L. A., Sawyer, J. E., Barker, D. W., Dietzel, R., Poffenbarger, H., Castellano, M. J., et al. (2016). Modeling long-term corn yield response to nitrogen rate and crop rotation, 7, 1–18. https://doi.org/10.3389/fpls.2016.01630
- Sacks, W. J., Deryng, D., Foley, J. A., & Ramankutty, N. (2010). Crop planting dates: An analysis of global patterns. *Global Ecology and Biogeography*, 19(5), 607–620. https://doi.org/10.1111/j.1466-8238.2010.00551.x
- Sellers, P. J., Los, S. O., Tucker, C. J., Justice, C. O., Dazlich, D. A., Collatz, G. J., & Randall, D. A. (1996). A revised land surface parameterization (SiB2) for atmospheric GCMs. Part II: The generation of global fields of terrestrial biophysical parameters from satellite data. *Journal of Climate*, 706–737. https://doi.org/10.1175/1520-0442(1996)009<0706:ARLSPF>2.0.CO;2
- Shangguan, W., Dai, Y., Duan, Q., Liu, B., & Yuan, H. (2014). A global soil data set for earth system modeling. Journal of Advances in Modeling Earth Systems. https://doi.org/10.1002/2013MS000293
- Sharma, P., & Jain, A. K. (2018). Assessing the effects of land use changes, nitrogen fertilizer application and deposition on nitrogen leaching in South and South East Asia. AGU Fall Meeting.
- Stehfest, E., Heistermann, M., Priess, J. A., Ojima, D. S., & Alcamo, J. (2007). Simulation of global crop production with the ecosystem model DayCent. *Ecological Modelling*, 209(2–4), 203–219. https://doi.org/10.1016/j.ecolmodel.2007.06.028
- Stone, K. C., Camp, C. R., Sadler, E. J., Evans, D. E., & Millen, J. A. (2010). C y r n f i s c p, 26(3), 429-438.
- Twine, T. E., Bryant, J. J., Richter, T. K., Bernacchi, C. J., Mcconnaughay, K. D., Morris, S. J., & Leakey, A. D. B. (2013). Impacts of elevated CO₂ concentration on the productivity and surface energy budget of the soybean and maize agroecosystem in the Midwest USA. *Global Change Biology*, 19(9), 2838–2852. https://doi.org/10.1111/gcb.12270
- United States Department of Agriculture. (2019). Natural resources conservation service. Soil survey geographic (SSURGO) database. Retrieved from https://sdmdataaccess.sc.egov.usda.gov
- Vanloocke, A., Bernacchi, C. J., & Twine, T. E. (2010). The impacts of Miscanthus×giganteus production on the midwest US hydrologic cycle. *GCB Bioenergy*. https://doi.org/10.1111/j.1757-1707.2010.01053.x
- Waha, K., Van Bussel, L. G. J., Müller, C., & Bondeau, A. (2012). Climate-driven simulation of global crop sowing dates. Global Ecology and Biogeography, 21(2), 247–259. https://doi.org/10.1111/j.1466-8238.2011.00678.x